CSC2401F - Complexity Theory Fall 1997

Here follows reference texts that I recommend for CSC 364 and would therefore be good introductory references for CSC 2401.

The Lecture Notes for 364 are available at the Bookstore. They are based on material taken from the following references

ffl \Introduction to Algorithms", T. Cormen, C. Leiserson and R. Rivest, MIT Press. ffl \Fundamentals of Algorithmics", G. Grassard and P. Bratley, Prentiss Hall. ffl \Randomized Algorithms", R. Motwani and P. Raghavan", Cambridge University Press.

ffl \A Programming Approach to Computability", A. Kfoury, R. Moll and M. Arbib, Springer Verlag.

Some other textbooks that can be used as references

ffl \Introduction to the Theory of Computing", M. Sipser, PWS Publishing. ffl \Computers and Intractability", M. Garey and D. Johnson, W.H. Freeman and Company.